

Temi di discussione

(Working Papers)

Intergenerational transmission in regulated professions and the role of familism

by Omar Bamieh and Andrea Cintolesi







Temi di discussione

(Working Papers)

Intergenerational transmission in regulated professions and the role of familism

by Omar Bamieh and Andrea Cintolesi

Number 1350 - October 2021

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Di Cesare, Raffaela Giordano, Monica Andini, Marco Bottone, Luca Citino, Francesco Franceschi, Danilo Liberati, Salvatore Lo Bello, Juho Taneli Makinen, Alessandro Moro, Tommaso Orlando, Claudia Pacella, Fabio Piersanti, Marco Savegnago, Stefania Villa. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

INTERGENERATIONAL TRANSMISSION IN REGULATED PROFESSIONS AND THE ROLE OF FAMILISM

by Omar Bamieh^{*} and Andrea Cintolesi^{*}

Abstract

We measure the extent to which familism accounts for the intergenerational transmission of jobs in regulated professions. Before 2004, local committees graded the Italian bar exams for lawyers, but since 2004, exams have been randomly assigned to external committees for grading. We proxy for family ties with the number of successful candidates sharing a family name and law firm address with a lawyer that is already registered. We estimate that the number of new entrants with a family tie drops by at least 10 per cent, while the number of new lawyers does not change, showing that familism accounts for an important part of the intergenerational transmission in our setting. While we do not find any significant differences by gender, familism is stronger in areas with low social capital, which also feature lower rents from licences.

JEL Classification: J44, J62. Keywords: lawyers, regulated professions, familism. DOI: 10.32057/0.TD.2021.1350

Contents

1.	Introduction	5
2.	Related literature	7
3.	Institutional background, data and family ties	10
	3.1 Institutional background	10
	3.2 Data and family ties	11
4.	Empirical strategy	12
5.	Main results	13
	5.1 Heterogeneities	19
	5.2 Robustness	20
	5.3 Where to target monitoring efforts: rents or social capital?	20
6.	Conclusions	25
Re	ferences	. 28
Ap	opendix: definitions of financially constrained firms	31
-	-	

^{*} University of Vienna, Economics Department.

^{**} Bank of Italy, Firenze Branch.

1 Introduction

Many professions are characterized by a licensing exam graded by a local committee, which is intended to be an important form of regulation for an industry to ensure quality standards. At the same time, despite the presence of licensing exams, intergenerational transmission in these professions remains sizable. For example, in the United States, approximately 10% of the members of regulated professions such as law and medicine have a parent who worked in the same profession (Laband and Lentz (1992); Lentz and Laband (1989)), and in Italy, the probability for a graduate's child to become a liberal professional doubles when the father is a graduate liberal professional (Aina and Nicoletti (2018)).

There are different environmental and economic factors that may explain the intergenerational transmission in regulated professions, such as family upbringing, genetic ability or regulations.¹ Beyond these factors, intergenerational transmission may also be related to familism, i.e. behaviours aimed at creating unfair advantages for relatives to enter the profession. Recent empirical evidence suggest that familism triggers negative selection (e.g. Basso et al. (2021); Durante, Labartino and Perotti (2011)),² raising the need to understand the extent to which familism accounts for intergenerational transmission. Besides, to design effective policy limiting familism, we need to understand which are the channels through which familism operates. At present, the empirical evidence on this subject is surprisingly scant.³ This paper fills this gap and identifies the role played by family connections, strength of social norms and economic incentives in creating dynasties in regulated professions.

Disentangling the share of intergenerational transmission generated by familism from other factors is empirically challenging and requires a suitable testing ground. For this reason, we focus on the legal profession in Italy, in which the licensing exam was reformed in 2004. Before 2004, the Italian bar exam was graded by a local committee, potentially leaving room for relatives to interfere with the grading process. In this respect, it was very similar to many current licensing exams across the world.⁴ Then, a reform severely limited

¹e.g., Björklund, Lindahl and Plug (2006); Aina and Nicoletti (2018); Mocetti (2016); Mocetti, Roma and Rubolino (2020).

 $^{^{2}}$ A positive selection could arise if the lower ability of new entrants is compensated by parental spillovers.

³An exception is Aina and Nicoletti (2018), who used a set of controls and concluded that familism seems to help with completing compulsory practice but not with passing the licensing exam.

⁴For example, in France, the bar exam is graded at the university level, and in Austria and Germany, it takes place at the state level. The populations of these jurisdictions are comparable to those of the Italian districts.

the extent to which parents can resort to unfair practices to increase the probability of their children passing the bar exam. Starting in 2004, the exams of each district are graded by an external committee that is randomly selected every year.⁵

We proxy for family ties by using the number of successful candidates with the same family name and law firm address of an already registered lawyer. Compared to the existing literature, (Güell, Mora and Telmer (2015)), using law firm addresses does not force us to rely on rare family names to identify plausible family linkages. Since 2004, the number of new entrants per inhabitant with family ties has declined by 10-30%. The number of lawyers did not change following the 2004 reform, suggesting that good candidates used to be crowed out by candidates with family ties. There are different reasons to interpret the estimates as lower bounds of the real effect of complete decentralization of grading committees: first, the 2004 reform decentralized only the written part, while the oral part still takes place at local level; second, we do not track family ties between relative with different surnames, like mother and child; third, family networks might well go beyond districts. To get a sense of the size of the estimated effects, note that since 2004, 4118 new lawyers entered the register with a family connection, representing approximately 4 percent of the new entrants. We estimate that, in the absence of the reform, approximately 1297 more lawyers with family ties would have been in the register. This effect is similar for males and females aspiring lawyers.

Lawyers entering the profession merely because they received family help lowers the standard of the profession (Basso et al. (2021)). This calls for policies intended to combat familism, and in the second part of the paper, we provide insights into where familism prevails. Evidence from the literature suggests that familism should be more common where the rents from a license are high (e.g., Brollo et al. (2013)) and social capital is low (e.g., Durante, Labartino and Perotti (2011)). We investigate whether familism correlates with rents from the license and/or social capital. This investigation is particularly interesting in our setting because the income of Italian lawyers is heterogeneous across geographical areas: districts with higher returns from the profession also have higher degrees of social capital, whereas districts with lower returns have lower social capital, making rents and social capital positively correlated. We repeat our empirical exercise for each Italian region, and we find that places with low social capital are those where familism practices are pervasive, even if the returns from the license are relatively lower

⁵Note that the reform may reduce familism because external committees are less likely to favour candidates with relatives in the profession, but also because relatives may not even try the exam if they know that they will not receive any help. The objective of this paper is to measure the overall effect and not the two channels separately.

than in other places.

The contribution of this paper is twofold. First, it disentangles the importance of familism in intergenerational transmission in regulated professions from other possible drivers. When committees can be reached through family ties, we estimate that approximately 10–30 percent of new entrants with a family tie pass the exam only because of family help. Given that most licensing committees in Western countries are as close to candidates' families as the ones we analyze, this result may be important for the correct design of current institutions controlling access to regulated professions. Second, this paper provides evidence that familism is pervasive in areas with low social capital, despite the lower rents from licenses, suggesting that beyond economic incentives other components may matter in spreading familism.

The rest of this paper is organized as follows: Section 2 reviews the literature and describes our measure of family ties. Section 3 explains the institutional settings of the Italian bar exam and the 2004 reform. Section 4 describes our empirical strategy. Section 5 shows how the 2004 reform affected familism and Section 6 concludes.

2 Related literature

This paper is related to four streams in the literature. First, it relates to the literature on familism and nepotism. There are two papers on this subject in the context of the legal profession. Basso et al. (2011) found a negative relationship between the age when people start a law practice in Italy and the frequency of their family name in the local register. This association decreased after 2004, pointing in the same direction as this study. Laband and Lentz (1992) studied the intergenerational transmission of the legal profession in the United States, and they concluded that the presence of nepotism at the entry stage cannot be rejected. Basso et al. (2011) and Laband and Lentz (1992) are interested in showing the presence of familism, instead our empirical exercise is primarily intended to quantify the relevance of it with respect to other drivers in accounting for intergenerational transmission. In our setting familism takes the form of cheating in a licensing exam, which can be seen as a form of corruption. Using data for Italy, Gagliarducci and Manacorda (2020), show that politicians extract significant rents, in terms of private sector jobs, for their family members.

Second, this paper relates to the literature on the intergenerational transmission of jobs. This literature stream has mostly focused on the role of innate ability versus that of family upbringing and environmental factors⁶ (Bowles, Gintis and Groves (2009); Björklund, Lindahl and Plug (2006); Sacerdote (2011); Lo Bello and Morchio (2020)). More recently, Mocetti (2016) showed that children are more likely to follow their parents' careers if positional rents favor their parents' profession. Aina and Nicoletti (2018) showed that there is a significant effect of familism on the probability of completing compulsory practice but not on passing the licensing exam. Unlike Aina and Nicoletti (2018), we do not rely on a set of controls to isolate the effect of familism but rather exploit exogenous variation. Furthermore, our sample is larger, as it contains all lawyers and not just those who graduated from a subset of universities. Beyond the previously studied channels, this paper contributes to the literature by showing that dynasties in professions are in part generated by familism.

Third, this paper is in line with the literature on the role of relatives in family firms (e.g., Bertrand and Schoar (2006)), which supports the idea that a culture based on strong family ties may sometimes impede economic development. Suboptimal economic organizations can emerge when relatives place excessive weight on keeping the business in the family, perhaps due to a strong sense of duty towards other family members or a more selfish desire to turn the business into a family legacy.⁷ In this paper, we show that senior lawyers not only help their relatives pass the bar exam but also hire relatives to work at their law firms. This familism may represent a cost because the relatives who passed the bar exam thanks to their family name are presumably less qualified than the average candidate. Therefore, senior lawyers gladly bear the cost of hiring less qualified junior lawyers to help family members.

Finally, this paper relates to the literature on social capital and social norms. Durante, Labartino and Perotti (2011) found that decentralization led to a significant increase in familism among Italian academics in areas with low social capital, but to a reduction in familism in areas with high social capital. Their empirical exercise regarded a public sector job, for which one can hire an unqualified relative without bearing any consequence. Compared to Durante, Labartino and Perotti (2011), this study focuses on a private sector job, a context where family members are more likely to internalize the costs of having an unqualified person working for them. Another difference relative to Durante, Labartino and Perotti (2011) is that the returns from accessing a public sector profession are higher in less developed areas of the country, a consequence of the equalized salaries in the Italian public sector. Given that less developed areas are also those with low social capital, one

⁶These factors might favor the transmission of occupation-specific human capital within the family and, therefore, intergenerational persistence in professions.

⁷Other contributions on the role of relatives in family firms include Bassanini et al. (2013).

cannot say whether the economic motive to help relatives in the academic profession prevails over the civic motive.

Family names have been used before to study intergenerational mobility. Güell, Mora and Telmer (2015) study intergenerational mobility using the joint distribution of surnames and economic outcomes. Their methodology relies on rare family names because only rare family names are indicative of familial linkages. Applying the same methodology to Italy, Güell et al. (2018) find that higher intergenerational mobility is positively associated with economic activity, education and social capital and negatively correlated with inequality. Our approach circumvents this problem because the Italian national register of lawyers contains information not only on family names but also law firm address. Italian law firms are usually very small, (the median number of lawyers working at the same law firm address is 3, and the average 6.95), therefore, two lawyers sharing their family name and working at the same law firm are very likely to be related. Using the law firm address as additional information to establish family links is particularly important in the Italian setting because the distribution of family names is more concentrated in the south than in the north of Italy.

Differently from Güell, Mora and Telmer (2015), we are not interested in identifying relatives versus nonrelatives in the population of lawyers. Instead, we only want to study if the share of relatives changes following the 2004 reform. In this respect, our paper is closer to Mocetti (2016); Mocetti, Roma and Rubolino (2020), who study how rents and the strictness of professional services regulation determine the propensity to follow the parents' career.

Similarly to our study, Gagliarducci and Manacorda (2020) define families as groups of individuals sharing the same first three consonants and born in the same municipality. Durante, Labartino and Perotti (2011), instead, use the presence of multiple professors with the same last name in the same department as a proxy for the presence of relatives. Fafchamps and Labonne (2017) study political connections in the Philippines, exploiting the fact that in each municipality, a particular family name was given to only one family. Santavirta and Stuhler (2020) compare different name-based estimators used in intergenerational research and show that their interpretation crucially depends on the sampling properties of the data.

3 Institutional background, data and family ties

3.1 Institutional background

The Consiglio Nazionale Forense (CNF) is the institution representing Italian lawyers. Among its functions, the CNF maintains the national register of lawyers, appoints the examiners for the bar exam that grants admittance to the profession, administers the elections of district committees, and resolves disputes and complaints regarding register membership and disciplinary measures decided by district committees.

A law degree is a prerequisite to becoming a lawyer in Italy.⁸ After graduating, the prospective lawyer begins a legal traineeship under the supervision of a senior lawyer. Upon completing the traineeship, the candidate can take the bar exam to become a lawyer. After successfully passing the bar exam, the new lawyer is registered in the lawyers' national register and can practice as a lawyer.

The Italian bar exam is composed of a written exam and an oral exam. Both exams take place in all 26 appellate courts. Candidates are assigned to the courts in the area where they had their traineeship. The written exam is anonymous and takes place in different locations simultaneously. The questions are identical across locations and are prepared in advance by the Ministry of Justice. Before 2004, the exams were graded by a local committee composed of lawyers, judges and law professors from the same geographical area in which the candidate took the exam. Grading standards varied significantly across districts, and pass rates were as low as 16% and as high as 96%(Pagliero and Buonanno (2018)). In the early 2000s, the enormous differences in pass rates across districts and anecdotal cases of the more lenient behavior of examiners towards relatives and friends raised concerns about the fairness of the bar exam. The Italian government responded by changing the system in 2004.⁹ Starting with the 2004 exam, all the written exams in a district must be graded by a committee from another district that is randomly selected every year. The reform only affected the procedure for grading of the written exam. The oral exam is still graded by a committee from the same district of the candidates.

The number of graders in each district depends on the number of candidates in that district. To avoid excessive workloads on graders, the random assignment of grading

⁸Law schools offer two degree levels. Students obtain the first degree by attending a three-year program and fulfilling all its requirements. Then, they can enroll in a more specialized two-year program. To become a lawyer, graduation from the five-year (full) program is required to sit for the bar exam.

⁹The so called Castelli reform, Law 180/2003, http://www.camera.it/parlam/leggi/03180l.htm

committees occurs within groups of districts that change marginally over time and consist of districts with similar numbers of candidates. For example, Milan, Rome, and Naples being the three most populous cities, have always been in the same group. Each year the ministry randomly draws Milan to correct the written exams from Rome or Naples, and the same happens to Rome and Naples.

3.2 Data and family ties

The CNF provided us with a full snapshot of the Italian register of lawyers as of 1 July 2017. It contains the following information for 205,311 lawyers: encrypted name, family name,¹⁰ location, gender, date of birth and date of registration in the CNF archives. Lawyers cannot practice unless they are registered in the CNF register. Municipal tax revenues per capita and fiscal capacity data come from the Ministry of the Interior.

Our data allow us to identify lawyers with the same family name, the same law firm address as of July 2017 and the year in which they entered the register. We measure the family ties of new entrants in each Italian municipality with two measures. First, the number of new entrants sharing the family name and law firm address as an already registered lawyer; for year t and municipality c we label this FA_{ct} . Second, we compute the number of new entrants sharing a family name with a lawyer already registered in the same municipality, and we label this FM_{ct} .

Both measures are defined per 10000 inhabitants in year t and municipality c. The first measure is more precise, but also narrower, than the second measure. On the one hand, lawyers with the same family name working in the same place are very likely to be relatives. On the other hand, lawyers with the same family name operating in the same municipality include relatives working elsewhere. Relatives working at different law firms are not rare, and the results show that familism towards them is as large as that for relatives who decide to work together after passing the exam.

Table 1 reports some summary statistics for FA_{ct} and FM_{ct} . Comparison of FA_{ct} and FM_{ct} between different territories may well capture differences in the underlying distribution of family names. The high number of lawyers sharing family names in a given region might be because there are few family names in that region. Our empirical strategy, described in the next section, carefully circumvents this issue because we look at the within-municipality variations of FA_{ct} and FM_{ct} at the moment of the reform, and it is unlikely that the distribution of surnames within-municipalities drastically changed

¹⁰Encryption does not allow us to see the real name and surname, but it does allow us to identify which lawyers have the same family name.

between 2004 and 2005. We discuss this in details in the following section.

	FA_{ct}			FM_{ct}					
	200	2000-2004		2005-2010		2000-2004		2005-2010	
Region	Mean	St. Dev.	Mean	St. Dev.	Mean	St. Dev.	Mean	St. Dev.	
Abruzzo e Molise	0.014	0.114	0.014	0.114	0.052	0.260	0.017	0.108	
Basilicata	0.017	0.170	0.040	0.432	0.101	0.585	0.092	0.487	
Calabria	0.034	0.254	0.028	0.264	0.226	0.762	0.100	0.420	
Campania	0.035	0.299	0.035	0.301	0.222	0.802	0.218	0.758	
Emilia Romagna	0.010	0.064	0.012	0.081	0.032	0.143	0.049	0.199	
Lazio	0.018	0.123	0.034	0.222	0.117	0.438	0.096	0.426	
Liguria	0.016	0.105	0.013	0.084	0.026	0.111	0.043	0.175	
Lombardia	0.010	0.080	0.010	0.085	0.032	0.210	0.032	0.227	
Marche and Umbria	0.018	0.108	0.015	0.089	0.048	0.183	0.046	0.188	
Piemonte and VdA	0.009	0.068	0.008	0.071	0.030	0.169	0.025	0.132	
Puglia	0.029	0.168	0.038	0.213	0.192	0.778	0.144	0.683	
Sardegna	0.016	0.242	0.017	0.127	0.072	0.378	0.064	0.355	
Sicilia	0.025	0.186	0.025	0.157	0.117	0.416	0.080	0.301	
Toscana	0.022	0.122	0.027	0.155	0.082	0.294	0.072	0.253	
Triveneto	0.012	0.097	0.014	0.138	0.019	0.133	0.019	0.125	
Italy	0.020	0.172	0.021	0.19	0.10	0.48	0.079	0.40	

Table 1: Descriptive statistics

Note.—Summary statistics for FA_{ct} and FM_{ct} between 2000 and 2004 and between 2005 and 2010. VdA means Valle d'Aosta. Trivento is formed by Friuli-Venezia Giulia, Veneto and the province of Trento. The province of Bolzano is excluded because it is the only province not affected by the reform.

4 Empirical strategy

We estimate a parametric event study model to learn the impact of the 2004 reform on familism. For municipality c and time t,¹¹

$$y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct}$$
(1)

where y_{ct} is FA_{ct} or FM_{ct} , $Reform_t$ is a dummy variable taking a value 1 for all years from 2005 (inclusive) onwards, f(t) is a time trend that is allowed to differ before and after 2005, X_{ct} includes time-varying municipality controls and ϕ_c represents the municipality fixed effects. The municipality fixed effect makes our comparison a within-municipality comparison, addressing concerns of heterogeneous family name distributions across municipalities.¹² Our coefficient of interest is β , which identifies the effect of the reform on

¹¹Appendix B presents results for a nonparametric event study.

¹²Note that to identify this effect, the municipality fixed effect controls for the heterogeneity in the distribution of family names in equation (1) under the plausible assumption that the heterogeneity is

the number of successful candidates sharing a family name and law firm address with a lawyer already registered in town c and year t. In other words, because the reform represents a permanent shock affecting the scope to which lawyers can influence exam committees, our estimates disentangle the discontinuous change at municipal level, from possible time trends, in namesake. Note that we exploit FA_{ct} and FM_{ct} to gauge the effect of the 2004 reform on familism and not to estimate the evolution of familism *over* time. Indeed, it is crucial to include a time trend in the empirical analysis: FA_{ct} and FM_{ct} have a time trend, which is generated by a the structure of the data and is fully explained in Appendix F, and a direct interpretation of them as temporal measures of familism may be misleading.

The exercise assumes that the year of registration is the year in which a lawyer passes the bar exam. Because lawyers cannot practice until they are registered, they have no reason to delay their registration once they have passed the bar exam.¹³ Therefore, the registration year is a reasonable proxy for the year in which a lawyer passed the bar exam.

Finally, because we seek to identify geographical heterogeneities, for every Italian region r composed of a set of municipalities M_r , we run the same but region-specific model:

$$y_{ct} = \alpha + \beta_r \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct} \quad \text{if} \quad c \in M_r \tag{2}$$

where the β_r is now a region-specific coefficient and the different β s allow us to rank regions according to their pre-reform familism levels.

5 Main results

Figure 1 plots the evolution of FA_{ct} (top panel), number of entrants with the same family name and law firm address of another lawyer, and FM_{ct} (bottom panel), number of entrants with the same family name of an already registered lawyer in the same municipality. Both variables are detrended over time. The number of successful candidates we proxy to be related to an already registered lawyer drops starting in 2005 as the first cohort of successful candidates under the new bar exam rules became lawyers.¹⁴ We interpret this result as evidence of familism because before the reform, written exams were graded locally, and senior lawyers could use their professional connections to provide easier access

constant over time.

¹³The results of the exam are published in June. So, it is unlikely that a lawyer forgoes half of his/her income to avoid yearly fix costs like registration or insurance costs. Further, this should cause stronger results to be present in areas with higher rents, but results in Figure 5 show the opposite.

¹⁴The first written exam under the new rules took place in 2004, but the oral exam is always scheduled during the next calendar year. Therefore, the first new entrants who took the exam under the new rules entered the profession in 2005.



Figure 1: Drop in family ties of new entrants at the time of reform

Note.—Time series for FA_{ct} and FM_{ct} . We estimate a linear trend in the pre-reform period, and we predict FA_{ct} for all the periods. The figure reports the difference between the observed data and the predicted time trend.

for their relatives taking the bar exam. Instead, this estimate does not capture the help a senior relative can provide to a junior candidate studying for the bar exam because the impact of candidate preparation on the exam does not depend on where the exam is graded. We interpret the drop in Figure 1 as a measure of how many lawyers passed the exam before the reform because they received help through their family connections. Those who received help before the reform but would have passed the exam anyway are not part of the drop.

To confirm the results of Figure 1, Table 2 reports the results for specification (1). In the upper panel, we use FA_{ct} as the dependent variable. All the specifications show a large decrease in the number of successful candidates sharing a family name and law firm address with an already registered lawyer. The estimates range from 0.006 to 0.008 percentage points when a linear trend is assumed, corresponding to a decline of approximately 20–30% at the mean value of the dependent variable in the year before the reform. The results are robust to the specification of a quadratic trend. The lower panel of Table 2 reports the results from specification (1), where the outcome variable is FM_{ct} , number of entrants with the same family name as another lawyer in the same municipality. The results are similar as those obtained with FA_{ct} , correspond to a decline of approximately 10% at the mean value of the dependent variable, and indicate that a significant share of the candidates who receive help from a family member do not then work with one, at least in the long run.¹⁵ Tables A1 and A2 in the appendix show further robustness to the specification of different time trends respectively for FA_{ct} and FM_{ct}

To understand how many lawyers would have entered the register with family help after 2004 in the absence of the reform, we perform a counterfactual exercise. We estimate a linear trend in FA_{ct} and FM_{ct} before the reform, and we extrapolate it for the years after the reform. In this way, we obtain a counterfactual for the fraction of new entrants sharing family names and law firm addresses had the reform not been implemented. Between 2004 and 2016, 98,576 new lawyers entered the profession. Among them, 4118 lawyers shared family names and law firm addresses with another lawyer already in the register. We estimate that in the absence of the reform, there would have been 1297 more of these lawyers. The actual number would have been 31% higher in the absence of the reduction in familism attributed to the reform.

¹⁵Since our data are a snapshot of the national register as of July 2017, we can observe only whether they share a law firm address in 2017, but we cannot observe whether they shared one previously.

Table 2: Reform effects on familism

(1)	(2)	(2)
(1)	(2)	(0)

A. Same family name and law firm address (FA_{ct})					
Mean of the dependent variable in	2004:	0.026.	Standard	deviation:	0.24.

Linear trend (N=29360)			
Reform	-0.006^{*} (0.003)	-0.006^{*} (0.003)	-0.008^{*} (0.004)
Quadratic trend $(N=95025)$			
Reform	-0.005^{*} (0.003)	-0.005^{*} (0.003)	-0.006 (0.006)

B. Same family name and municipality (FM_{ct})

Mean of the dependent variable in 2004: 0.10. Standard deviation: 0.48.

 Linear trend (N=29360)

 Reform
 -0.021^{**} -0.021^{**}

 (0.009)
 (0.009)

 Quadratic trend (N= 95025)

 Reform
 -0.013^{*}

 -0.013^{*}

	(0.007)	(0.007)	(0.011)
Municipality fixed effect	NO	YES	YES
Economic controls	NO	NO	YES

-0.022**

(0.009)

 -0.024^{**}

Note.—Standard errors clustered at the district level are in parentheses. * p < 0.1, ** p < 0.05. These are results for specification 1. ($y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct}$). Control variables: the time trend that is allowed to differ before and after the reform, the population in columns 2, 3 and 4 and the income per capita and fiscal capacity of the municipality in column 4. The time period considered is 2001-2008 when controlling for a linear trend, 1993-2016 when controlling for a quadratic trend and 1998-2016 when including economic controls and a quadratic trend.

Table 3: Heterogeneous effects of the reform with respect to gender

	Male							
	(1)	(2)	(3)	(4)	(5)	(6)		
A. Same family name and law firm address (FA_{ct}) Mean of the dependent variable in 2004: male=0.011, female=0.008. Standard deviation: male=0.19, female=0.10.								
Linear trend (N=29224)								
Reform	-0.003 (0.002)	-0.003 (0.002)	-0.004^{*} (0.002)	-0.002 (0.002)	-0.002 (0.002)	-0.002 (0.002)		
Quadratic trend (N= 94	583)							
Reform	-0.002 (0.002)	-0.002 (0.002)	-0.002 (0.004)	-0.002 (0.002)	-0.002 (0.002)	-0.003 (0.002)		
B. Same family name a Mean of the dependent var Standard deviation: male=	nd muni riable in 2 =0.30, fem	cipality 004: male ale=0.22	(<i>FM_{ct}</i>) e=0.038, fe	male=0.03	1.			
Linear trend (N=29224)								
Reform	-0.009^{*} (0.005)	-0.009^{*} (0.005)	-0.010^{*} (0.005)	-0.004 (0.005)	-0.004 (0.005)	-0.003 (0.005)		
Quadratic trend (N= 94	Quadratic trend $(N=94583)$							
Reform	-0.008* (0.004)	-0.008* (0.004)	-0.010 (0.006)	-0.002 (0.004)	-0.002 (0.004)	-0.006 (0.006)		
Municipality fixed effect Economic controls	NO NO	YES NO	YES YES	NO NO	YES NO	YES YES		

Note.—Standard errors clustered at the district level are in parentheses. * p < 0.1. These are results for specification 1. $(y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct})$. Control variables: the time trend that is allowed to differ before and after the reform, the population in columns 2, 3 and 4 and the income per capita and fiscal capacity of the municipality in column 4. The time period considered is 2001-2008 when controlling for a linear trend, 1993-2016 when controlling for a quadratic trend and 1998-2016 when including economic controls and a quadratic trend.

	(1)	(2)	(3)					
A. New Entrants per 10000 Inhabitants								
Mean of the dependent variable in 2004: 0.81. Standard deviation: 1.84.								
Linear trend (N=29360)								
Beform	-0.004	-0.004	0.012					
	(0.041)	(0.041)	(0.039)					
	()		()					
Quadratic trend $(N=95025)$								
5.4	0.010							
Reform	-0.013	-0.013	0.003					
	(0.047)	(0.047)	(0.065)					
B. Males: New Entrants per 10000 Inhabitants	5							
Mean of the dependent variable in 2004: 0.37. Standa	ard deviat	tion: 1.25						
Linear trend $(N-20360)$								
Emean trend (N=23500)								
Reform	-0.039	-0.039	-0.027					
	(0.027)	(0.027)	(0.026)					
	. ,	. ,	. ,					
Quadratic trend $(N=95025)$								
	0.004	0.004	0.007					
Reform	-0.034	-0.034	-0.037					
	(0.020)	(0.020)	(0.055)					
C. Females: New Entrants per 10000 Inhabitan	nts	• 1.01						
Mean of the dependent variable in 2004: 0.43. Standa	ard deviat	tion: 1.21						
Linear trend $(N-29360)$								
Reform	0.038^{*}	0.038^{*}	0.042^{*}					
	(0.022)	(0.022)	(0.021)					
Quadratic trend $(N = 95025)$								
Boform	0 023	0 023	0.040					
TWIOT III	(0.026)	(0.026)	(0.040)					
Municipality fixed affect	NO	VFC	VFC					
Economic controls	NO	I LO NO	I LO VES					
	nO	шU						

Table 4: Effect of the reform on the number of entrants

Note.—Standard errors clustered at the district level are in parentheses. * p < 0.1. These are results for specification 1. $(y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct})$. Control variables: the time trend that is allowed to differ before and after the reform, the population in columns 2, 3 and 4 and the income per capita and fiscal capacity of the municipality in column 4. The time period considered is 2001-2008 when controlling for a linear trend, 1993-2016 when controlling for a quadratic trend and 1998-2016 when including economic controls and a quadratic trend.

	Linear trend		Quadratic trend	d				
	(1)	(2)	(3)	(4)				
New entrants average age								
Mean of the dependent variable in 2004: 32.52. Standard deviation: 3.65.								
Reform	0.04	0.08	0.08	0.33				
	(0.126)	(0.159)	(0.138)	(0.214)				
Observations	26234	21316	26234	21316				
Municipality fixed effect	YES	YES	YES	YES				
Economic controls	NO	YES	NO	YES				

Table 5: Reform effects on age at entry into the profession

Note.—Standard errors clustered at the district level are in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01. These are results for specification 1. ($y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct}$). Control variables: the time trend that is allowed to differ before and after the reform, the population and, in columns 2 and 4, the income per capita and fiscal capacity of the municipality. The time period considered is 2001-2008 when controlling for a linear trend, 1993-2016 when controlling for a quadratic trend and 1998-2016 when including economic controls and a quadratic trend.

5.1 Heterogeneities

Having established that the number of relatives drops with the decentralization of the grading system, we study if there are heterogeneous effects with respect to gender and if the reform affected other lawyers' characteristics, such as the absolute number of new lawyers, their age, and gender composition.

Table 3 reports result from the same analysis as in table 2, for men and women separately. In columns (1), (2), and (3), the outcomes are FA_{ct} and FM_{ct} computed just for males. Columns (4), (5), and (6) instead look at female lawyers. Although the estimates are imprecise, the effect of the reform decreased the number of male familistic lawyers, more than the number of female familistic lawyers. However, the mean of the dependent variable for males is larger than for females, and the percentage drops are similar, suggesting that familism accounts for the same share of intergenerational trasmission in both groups.

We test if the reform changed the number of new lawyers estimating the same specification as before, equation (1), using the number of new lawyers as an outcome. Panel A of table 4 reports the results and shows that the reform did not change the number of new lawyers. Although there is no quota on the number of new lawyers, our evidence shows that exam committees might have a focal point of how many people can pass the bar exam every year. In fact, the oral part of the exam is still locally graded, hence this is where examiners can regulated the number of new entrants. When it was possible for senior lawyers to help their younger relatives, those relatives took the spot of nonrelatives in the bar exam, as documented in the previous section. Next, we test if the reform changed the gender composition of new lawyers. It could be that females, more than males, used to be discouraged by the unfair familist practices that plagued the bar exam and, because of this, did not even try to take the exam. Panel B shows that the reform did not affect the number of males new lawyers. Panel C shows instead that the number of female new lawyers increased by about 5-10% following the implementation of the reform. This evidence is in line with the results reported in Table 3. However, these estimates are imprecise and not robust to different specifications of the time trend.

Finally, we assess if the reform has changed the age at which people become lawyers. Law school graduates aspiring to become lawyers might fail in their first attempt at the bar exam. Presumably, the age at which one becomes a lawyer measures the number of attempts needed to pass the bar exam, which proxies candidates' quality. Table 5 shows that the reform had no effect on the profession's age at entry for lawyers.

5.2 Robustness

Figure 2 plots the estimations of a placebo test for the results in Table 2 that falsifies the year of the reform. The x-axis shows the false year of reform, and the y-axis shows the relative placebo coefficient and its 95% confidence interval. The figure on the left reports the results using all the pre-reform periods, while that on the right reports those for all the post-reform periods. Overall, the results are not significant and much smaller in size than the estimates in Table 2. Tables A1 and A2 confirm the robustness of our main findings with respect to different specification of the time trend.

Appendix E replicates the main results of the study using an alternative definition of the outcome variable. We divide the outcome variables used so far, the number of new entrants with the same family name and law firm address as an already registered lawyer (FA_{ct}) and the number of new entrants with the same family name as a lawyer already registered in the same municipality (FM_{ct}) , by the number of newly registered lawyers in year t and municipality c. This alternative definition of the outcome produce results qualitatively equivalent to those of the main section of this study.

5.3 Where to target monitoring efforts: rents or social capital?

The literature generally holds that whether intergenerational transmission is desirable is debatable (e.g., Bowles, Gintis and Groves (2009); Björklund, Lindahl and Plug (2006); Sacerdote (2011)); however, concerning a regulated profession, it may be inefficient if it

Figure 2: Placebo estimates



Note.—Placebo estimates from specification (1) for FA_{ct} (above) and FM_{ct} (below). In the left (right) panels, we run specification (1) $(y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct})$ using each year as the false year of the reform with all the pre-reform (post-reform) periods. The estimates are not significant and small in size, confirming the robustness of the results in Table 2.

relates to familism designed to favor the entrance of relatives in the profession. Clearly, our results follow this case and reveal inefficient behavior that lowers the standard in the profession. This consideration naturally raises another question, especially for a monitoring authority interested in reducing familism: where should we look for familism?

The existing literature has shown that misbehavior prevails where rents are higher (e.g., Brollo et al. (2013)) and social capital is lower (e.g., Durante, Labartino and Perotti (2011)). In our setting, this phenomenon is particularly interesting: we estimate rents from being a lawyer in every Italian region as the net salary markup over the average salary of a graduate person in the same region including those without an employment and correcting the salary of a lawyer with a measure of possible tax evasion.^{16 17} Figure 3 plots the correlation between rents from being a lawyer against different measures of social capital. Clearly, rents and social capital are positively correlated, generating a tradeoff in monitoring efforts for policymakers interested in reducing familism. Although the fact that rents are lower in Southern region could be surprising, it is consistent with the fact that in the South of Italy, regulated occupations represent a higher share of total employment than in the North (Mocetti, Rizzica and Roma (2021)): the higher competition in regulated occupation is one of the possible explanation for the lower level of rents in the South.

Table 6 reports the estimated familism for each region separately under specification (2). We prefer to use FA_{ct} as the dependent variable because it results in more precise estimates (it also considers familism towards relatives of lawyers who do not join the law firm of the senior lawyer, so it has greater variation), but the results are qualitatively similar using FM_{ct} , as reported in appendix D. In Figures 4 and 5, we plot the estimated degree of familism against measures of social capital and rents from being a lawyer. Figure 4 shows that lower levels of social capital are associated with higher familism, while Figure 5 suggests that rents are negatively associated with it. This finding may be attributed to the different levels of variation in social capital and rents. However, the variation in rents is remarkably large, as they range between the lowest level of 1,208 EUR in Calabria and the highest of 5,457 EURO in Lombardia, with a standard deviation of 1,331 EUR.

¹⁶An alternative comparison would consider the salary markup over a graduate from law school in 2003. Some versions of this type of data are available for only a few schools; they cover incomes for a few years after graduation, and they do not include the potential rents of a full career, which is particularly important in a profession such as lawyers in Italy for which returns at early stages of the career are very low (or null).

¹⁷Data for the average salary of a graduate are from the publicly available version of the Labour Force Survey. Given the lack of estimate on labour income tax evasion at regional level, we use a regional estimate of the tax evasion on property tax in 2012 produced by the Italian Revenue Agency as measure of tax evasion.

Region	Coefficient	Standard Error	p-value	Towns	Pre-Reform Average
Calabria	-0.121	0.043	0.00	323	0.23
Puglia	-0.082	0.028	0.00	221	0.19
Basilicata	-0.053	0.062	0.39	117	0.10
Campania	-0.046	0.039	0.24	453	0.22
Sicilia	-0.035	0.020	0.09	281	0.12
Lazio	-0.028	0.035	0.41	176	0.12
Sardegna	-0.010	0.024	0.68	157	0.07
Piemonte e VdA	-0.007	0.012	0.59	175	0.03
Abruzzo and Molise	0.003	0.017	0.86	204	0.05
Lombardia	0.004	0.007	0.55	707	0.03
Triveneto	0.008	0.011	0.45	177	0.02
Toscana	0.010	0.021	0.64	190	0.08
Emilia Romagna	0.010	0.008	0.21	235	0.03
Marche and Umbria	0.017	0.014	0.25	169	0.05
Liguria	0.020	0.018	0.29	73	0.03

Table 6: Regional estimates of familism

Note.— We run specification (2) for town c in region d: $FM_{ct} = \alpha + \beta_d \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct}$ if $c \in d$. For each region, column 1 reports the estimated β , column 2 its standard error, column 3 its p-value and column 4 the number of towns used in the estimation (i.e., those that have registered lawyers). We merge some geographically nearby regions because they do not have enough towns to provide an adequate level of precision in the estimates. For the same reason, we prefer FM_{ct} to FA_{ct} , although the results are qualitatively the same. Then, we merge Piemonte and Valle d'Aosta, Abruzzo and Molise, Marche and Umbria, Veneto and Friuli-Venezia Giulia and province of Trento. The reform did not change the rules in the province of Bolzano.



Figure 3: Social capital and rents from the license

Note.—Correlation of social capital and rents from being a lawyer across Italian regions. We estimate rents from being a lawyer in every Italian region as the gross salary markup over the average salary of a graduate person in the same region including unemployment and a correction for the estimate tax evasion in the region. We measure social capital with the most commonly used indicator of social capital in the literature for Italy: levels of newspaper readership, levels of blood donation, political turnout in the divorce referendum and estimated cheating on the Istituto Nazionale per la Valutazione del Sistema Educativo di Istruzione e di Formazione (INVALSI) test in elementary school (sources: Cartocci (2007); Guiso, Sapienza and Zingales (2004) and Invalsi). Overall, lower social capital is associated with lower rents from the license.



Figure 4: Social capital and familism

Note.— Correlations between measures of social capital and familism across Italian regions. We consider the indicators of social capital most commonly used in the literature: levels of blood donations, political turnout, estimated cheating on the Invalsi test at elementary school and newspaper readership. Familism for region r is measured as $-\beta_r$ from specification 2. Overall, familism is higher in areas with lower levels of social capital.

6 Conclusions

The intergenerational transmission of jobs between relatives is common across countries (Corak and Piraino (2011); Lentz and Laband (1990)), and it is generated by different factors (e.g., Björklund, Lindahl and Plug (2006); Aina and Nicoletti (2018); Mocetti (2016); Mocetti, Roma and Rubolino (2020)). In regulated professions, an additional driver of intergenerational transmission is familism, i.e., the misbehavior of people already in a profession intended to favor relatives on the licensing exam.

In this paper, we measure the extent to which familism accounts for intergenerational transmission in regulated professions. We examine the effects of the 2004 reform of the Italian bar exam. Before 2004, local committees graded the exams, whereas from 2004 onwards, exams were randomly assigned to external committees. We find that in Italy, approximately 10–30 percent new lawyers with a family tie entered the register with



Figure 5: Rents from the license and familism

Note.— Correlations between rents from being a lawyer and familism across Italian regions. We estimate rents from being a lawyer in every Italian region with four different measures. In the top right panel we consider the gross salary of a lawyer, in the top left panel we correct the gross salary of a lawyer for the presumed tax evasion, in the bottom right panel we use the markup of the salary of a lawyer corrected for the presumed tax evasion over the average salary of a graduate person in the same region and in the bottom left we correct the latter measure considering also unemployed graduate. Overall, familism is higher in areas with lower levels of rents, suggesting that weak social norms are more important determinants of familism than economic incentives.

the help of a relative before 2004. Since 2005, approximately 4118 lawyers entered the profession and worked in the same law firm as a relative. We conclude that familism at the licensing stage is an important determinant of intergenerational transmission.

The presence of such a high level of familism is undesirable because new entrants who enter with the help of a family member lower the standards of the profession, and this phenomenon naturally calls for analysis of where familism should be sought. The literature suggests that low levels of social capital and high rents from a license may be two drivers of familism. In our setting, rents and social capital are positively correlated, generating a possible tradeoff for a policymaker seeking to reduce familism. We estimate the effect of the reform at regional level, and we find that familism is associated with lower levels of social capital, despite the fact that low social capital areas are those with lower rents from the license.

The results of this paper may be relevant in shaping institutions controlling access to regulated professions. However, care should be taken in generalizing the results to other countries: different levels of social capital, different strengths of family ties, different quality of institutions and a number of other factors can undermine the validity of the results across countries.

References

- Aina, Carmen, and Cheti Nicoletti. 2018. "The intergenerational transmission of liberal professions." *Labour Economics*.
- Bassanini, Andrea, Thomas Breda, Eve Caroli, and Antoine Rebérioux. 2013. "Working in family firms: Paid less but more secure? Evidence from French matched employer-employee data." *ILR Review*, 66(2): 433–466.
- Basso, Gaetano, Andrea Catania, Giovanna Labartino, Davide Malacrino, Paola Monti, and Michele Pellizzari. 2011. "Family ties in licensed professions in Italy." A report for the Fondazione Rodolfo Debenedetti, Milan.
- Basso, Gaetano, Eleonora Brandimarti, Michele Pellizzari, and Giovanni Pica. 2021. "Quality and selection in regulated professions."
- Bertrand, Marianne, and Antoinette Schoar. 2006. "The Role of Family in Family Firms." *Journal of Economic Perspectives*, 20(2): 73–96.
- Björklund, Anders, Mikael Lindahl, and Erik Plug. 2006. "The origins of intergenerational associations: Lessons from Swedish adoption data."
- Bowles, Samuel, Herbert Gintis, and Melissa Osborne Groves. 2009. Unequal chances: Family background and economic success.
- Brollo, Fernanda, Tommaso Nannicini, Roberto Perotti, and Guido Tabellini. 2013. "The political resource curse." *American Economic Review*, 103(5): 1759–96.
- **Cartocci, Roberto.** 2007. *Mappe del tesoro: atlante del capitale sociale in Italia*. Vol. 168, Il mulino.
- Corak, Miles, and Patrizio Piraino. 2011. "The intergenerational transmission of employers." *Journal of Labor Economics*, 29(1): 37–68.
- **Durante, Ruben, Giovanna Labartino, and Roberto Perotti.** 2011. "Academic Dynasties: Decentralization and Familism in the Italian Academia." National Bureau of Economic Research, Cambridge, MA.
- Fafchamps, Marcel, and Julien Labonne. 2017. "Do politicians' relatives get better jobs? evidence from municipal elections." The Journal of Law, Economics, and Organization, 33(2): 268–300.

- Gagliarducci, Stefano, and Marco Manacorda. 2020. "Politics in the family: Nepotism and the hiring decisions of Italian Firms." *American Economic Journal: Applied Economics*, 12(2): 67–95.
- Güell, Maia, José V.Rodríguez Mora, and Christopher I. Telmer. 2015. "The informational content of surnames, the evolution of intergenerational mobility, and assortative mating." *Review of Economic Studies*, 82(2): 693–735.
- Güell, Maia, Michele Pellizzari, Giovanni Pica, and José V. Rodríguez Mora. 2018. "Correlating Social Mobility and Economic Outcomes." *Economic Journal*, 128(612): F353–F403.
- Guiso, Luigi, Paola Sapienza, and Luigi Zingales. 2004. "The role of social capital in financial development." *American economic review*, 94(3): 526–556.
- Laband, David N, and Bernard F Lentz. 1992. "Self-recruitment in the legal profession." *Journal of Labor Economics*, 10(2): 182–201.
- Lentz, Bernard F, and David N Laband. 1989. "Why so many children of doctors become doctors: Nepotism vs. human capital transfers." *Journal of Human Resources*, 396–413.
- Lentz, Bernard F, and David N Laband. 1990. "Entrepreneurial success and occupational inheritance among proprietors." *Canadian Journal of Economics*, 563–579.
- Lo Bello, Salvatore, and Iacopo Morchio. 2020. "Like father, like son: Occupational choice, intergenerational persistence and misallocation." Discussion paper.
- Mocetti, Sauro. 2016. "Dynasties in professions and the role of rents and regulation: Evidence from Italian pharmacies." *Journal of Public Economics*, 133: 1–10.
- Mocetti, Sauro, Giacomo Roma, and Enrico Rubolino. 2020. "Knocking on parents' doors: regulation and intergenerational mobility." *Journal of Human Resources*.
- Mocetti, Sauro, Lucia Rizzica, and Giacomo Roma. 2021. "Regulated occupations in Italy: Extent and labour market effects." *International Review of Law and Economics*, 66: 105987.
- **Pagliero, Mario, and Paolo Buonanno.** 2018. "Occupational licensing, labor mobility, and the unfairness of entry standards."

- Sacerdote, Bruce. 2011. "Nature and nurture effects on children's outcomes: What have we learned from studies of twins and adoptees?" In *Handbook of Social Economics*.
- Santavirta, Torsten, and Jan Stuhler. 2020. "Name-Based Estimators of Intergenerational Mobility." Mimeo., Stockholm University.

A Robustness of the parametric event study

Table A1: Robustness to time trend specifications for FA_{ct}

Dep. Var.:Same family name and law firm address (FA_{ct}) Mean of the dependent variable in 2004: 0.026. St. deviation: 0.24.

(1) (2) (3)

Trend:	quadratic	before and	cubic	after	the reform	n (N=9502)	25)
Reform					-0.008^{**} (0.003)	-0.008** (0.003)	-0.009^{*} (0.005)

Trend: cubic before and quadratic after the reform (N=95025)

Reform	-0.007^{*}	-0.007^{*}	-0.012**
	(0.004)	(0.004)	(0.005)

Different cubic trends before and after the reform (N=95025)

Reform	-0.006	-0.006	-0.022**
	(0.004)	(0.004)	(0.009)

Different linear trends before and after the reform (N=80353) Time range: 9 years before/after the reform.

Reform	-0.006^{**} (0.002)	-0.005^{**} (0.002)	-0.004 (0.003)
Municipality Fixed Effect	NO	YES	YES
Economic Controls	NO	NO	YES

Note.—Standard errors clustered at the district level are in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01. These are results for specification 1. ($y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct}$). Control variables: the time trend, the population in columns 2, 3 and 4 and the income per capita and fiscal capacity of the municipality in column 4.

Table A2:	Robustness	to	time trend	specifications	for	FM_{ct}
-----------	------------	---------------------	------------	----------------	-----	-----------

Dep.	Var.:Sam	e family	v name	an	d law	[,] firm	add	lress (FM_{c}	ct)
Mean	of the deg	pendent [,]	variable	in	2004:	0.010.	St.	deviation:	0.48.

	(1)	(2)	(3)					
Trend: quadratic before and cubic after the reform $(N=94583)$								
Reform	-0.030*** (0.010)	-0.030^{***} (0.010)	-0.041^{***} (0.014)					
Trend: cubic before and quadratic after t	the reform	(N=94583))					
Reform	-0.035^{**} (0.014)	-0.035^{**} (0.014)	-0.056^{***} (0.017)					
Different cubic trends before and after the	ne reform ((N=94583)						
Reform	-0.030^{**} (0.013)	-0.030^{**} (0.013)	-0.077^{***} (0.021)					
Different linear trends before and after the reform (N=79979) Time range: 9 years before/after the reform.								

Reform	-0.005	-0.008	-0.007
	(0.005)	(0.005)	(0.007)
Municipality Fixed Effect	NO	YES	YES
Economic Controls	NO	NO	YES

Note.—Standard errors clustered at the district level are in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01. These are results for specification 1. ($y_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct}$). Control variables: the time trend, the population in columns 2, 3 and 4 and the income per capita and fiscal capacity of the municipality in column 4.

B Nonparametric event study

This section considers a more flexible nonparametric event study model. We analyze the coefficients on various indicator variables for time relative to the year of implementation of the reform (2004). The advantage of this approach is that it allows us to visually assess the pattern of outcomes relative to the year 2004. The basic specification takes the form,

$$y_{ct} = X_{ct}\gamma + \sum_{t=1999}^{2003} \mu_t + \sum_{t=2005}^{2016} \mu_t + \phi_c + \varepsilon_{ct}$$
(3)

where y_{ct} is detrended FA_{ct} or FM_{ct} , X_{ct} is a vector control variables, μ_t are coefficients on dummy for each year t and ϕ_c represents the municipality fixed effects. The set of coefficients μ_t for $t \neq 2004$ describe the evolution of y_{ct} before and after the reform with respect to the last year before the reform implementation.

Figure B1 shows the impact of the 2004 reform on our two measures, FA_{ct} and FM_{ct} , of family ties. For each outcome, we plot the estimated coefficients on event time (μ_t s) from the nonparametric event study regression (equation (3)).



Figure B1: Nonparametric event study

C Evolution of entrants over time



Figure C1: New entrants

Figure C2: New entrants with a family tie



D Ranking FA_{ct}

Region	Coefficient	Standard Error	p-value	Towns	Pre Reform Average
Campania	-0.027	0.026	0.29	453	0.06
Puglia	-0.023	0.013	0.07	221	0.04
Calabria	-0.027	0.011	0.04	323	0.04
Sicilia	-0.017	0.013	0.17	281	0.03
Sardegna	-0.010	0.013	0.51	157	0.02
Abruzzo and Molise	-0.005	0.007	0.53	204	0.02
Triveneto	0.002	0.010	0.83	177	0.02
Piemonte e VdA	0.001	0.010	0.96	175	0.02
Lombardia	0.001	0.003	0.85	707	0.01
Emilia Romagna	0.002	0.005	0.60	235	0.01
Liguria	0.003	0.007	0.62	73	0.01
Marche and Umbria	0.009	0.009	0.33	169	0.01
Lazio	0.009	0.011	0.43	176	0.02
Basilicata	0.010	0.037	0.77	117	0.03
Toscana	0.020	0.014	0.15	190	0.02

Table A3: Regional estimates of familism with FA_{ct}

Note.— We run specification (2) for town c in region d: $FA_{ct} = \alpha + \beta_d \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct}$ if $c \in d$. For each region, column 1 reports the estimated β , column 2 its standard error, column 3 its p-value and column 4 the number of towns used in the estimation (i.e., those that have registered lawyers). We merge some geographically nearby regions because they do not have enough towns to provide an adequate level of precision in the estimates. For the same reason, we prefer FM_{ct} to FA_{ct} , although the results are qualitatively the same. Then, we merge Piemonte and Valle d'Aosta, Abruzzo and Molise, Marche and Umbria, Veneto and Friuli-Venezia Giulia and province of Trento. The reform did not change the rules in the province of Bolzano.

E Alternative definition of the outcome

This appendix shows the robustness of our main findings with respect to alternative definitions of family ties.

E.1 Familist per new entrants

As for our main results, we measure the family ties of new entrants in each Italian municipality with two measures. First, we compute the fraction of new entrants sharing the family name and law firm address as an already registered lawyer. Second, we compute the fraction of new entrants sharing a family name with a lawyer already registered in the same municipality. Formally, our proxies for the family ties of new entrants in year t and municipality c are,

$$\tilde{FA}_{ct} = \frac{\text{new entrants with the same family name and law firm address as an already registered lawyer_{ct}}{\text{new entrants}_{ct}}$$

$$\tilde{FM}_{ct} = \frac{\text{new entrants with the same family name as a lawyer already registered in the same municipality}_{ct}}{\text{new entrants}_{ct}}$$

Compared to our main outcome measures, $\tilde{F}A_{ct}$ and $\tilde{F}M_{ct}$ condition on municipality c having at least one new registered lawyer in year t. Although this might be reasonable to capture the share of lawyers with family ties, it can mismeasure shifts in familism. For example, consider a hypothetical town where all new lawyers must have family ties to pass the bar exam, the reform changes this, and by making it harder to pass the bar exam without help from a relative, nobody in the same town passes the bar exam anymore. $\tilde{F}A_{ct}$ and $\tilde{F}M_{ct}$ would not capture this behavior because they are defined conditionally on new lawyers entering into the profession.

Table A4 present the main results using the alternative definitions of familism defined by \tilde{FA}_{ct} and \tilde{FM}_{ct} .

	(1)	(2)	(3)	(4)
Same family name and law firm add Mean of the dependent variable in 2003: (ress (\tilde{FA}_{ct}) 0.026. Stand) dard deviat	tion: 0.116.	
Linear trend				
Reform	-0.014^{***} (0.003)	-0.014^{***} (0.003)	-0.015^{***} (0.003)	-0.010^{**} (0.004)
Quadratic trend				
Reform	-0.007^{*} (0.004)	-0.007^{*} (0.004)	-0.007^{*} (0.004)	-0.010 (0.007)
Same family name and municipality Mean of the dependent variable in 2003: ((\tilde{FM}_{ct}) 0.103. Stand	dard deviat	tion: 0.217.	
Linear trend				
Reform	-0.021^{***} (0.007)	-0.021^{***} (0.007)	-0.017^{**} (0.008)	-0.032^{**} (0.012)
Quadratic trend				
Reform	-0.021^{***} (0.007)	-0.021^{***} (0.007)	-0.017^{**} (0.008)	-0.032^{**} (0.012)
Observations	26355	26355	26355	21412

				· · · ·		· · · ·
Table A4: 1	Robustness	of main	results:	FA_{ct}	and	FM_{ct}

Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01. These are results for specification 1. $(FA_{ct} = \alpha + \beta \cdot Reform_t + f(t) + \gamma X_{ct} + \phi_c + \varepsilon_{ct})$. Controls: time trend that is allowed to differ before and after the reform, population in columns 2,3 and 4 and income per capita and fiscal capacity of the municipality in column 4. New entrants with family ties decline drastically following the reform.

Municipality Fixed Effect

Economic Controls

NO

NO

NO

NO

YES

NO

YES

YES

F Measurement error and attrition

F.1 Measurement error

 FA_{ct} and FM_{ct} are proxies for family ties and may be affected by measurement error. FM_{ct} is less precise and includes lawyers that have the same surname and operate in the same municipalities but are not relatives. On the contrary, FA_{ct} is definitely more precise, but fails in measuring family ties between relatives that do not work together. Here we briefly discuss how those measurement errors may affect the results.

 FM_{ct} may produce false positives, i.e. it identifies family ties between lawyers with the same surname operating in the same municipalities that are not relatives. In this case, the measurement error does not influence the results if the number of false positives does not jump with the reform. Indeed, FM_{ct} can be decomposed as $FM_{ct} = t_{ct} + f_{ct}$ where t_{ct} are the true positives and f_{ct} the false positives. Our empirical exercise measures the variation in FM_{ct} at the time of the reform. Then, assuming that f_{ct} does not change discontinuously at the time of the reform, the variation in FM_{ct} is an unbiased estimate of the variation in t_{ct} .

 FA_{ct} may also produce false negatives, i.e. it does not identify family ties between relatives that do not work together. In this case, the measurement error affects the results if the effect of the reform on lawyers with a correctly specified family tie and lawyers who are false negatives is not the same. Admittedly, we cannot measure the number of false negatives and assess whether the reform affected them differently. However, note that the false negatives of the measure FA_{ct} are considered as lawyers with family ties when using FM_{ct} . Further, the results that we obtain using FM_{ct} are similar to those obtained with FA_{ct} : this suggests that the false negatives are either very few or receive the same effect due to the reform than those with a true family tie.

F.2 Attrition

As mentioned in the main text, our data are a snapshot of the Italian register of lawyers as of July 2017. This generates attrition on the variables FA_{ct} and FM_{ct} related to the structure of the data, and FA_{ct} and FM_{ct} cannot be interpreted as temporal measures of familism for at least three different reasons that may generate mechanical trends. Hereafter we report them, and then we explain under which conditions FA_{ct} and FM_{ct} can still be exploited to measure the variation in entrants with family ties at the time of the reform.

- 1. The growth in the number of lawyers that occurred in the 2000s may increase the chance of sharing a family name with another lawyer.
- 2. Because the data are a snapshot of the national register as of July 2017, the variables FA_{ct} and FM_{ct} may underestimate family ties the further we move away from 2017. For example, in year t, some senior lawyers may be active and contribute to the numerator of FA_{ct} and FM_{ct} that were no longer active in 2017. Unfortunately, we do not observe them in the data. The amount of these missing links increase the further t is from 2017, and the underestimation of the numerators of FA_{ct} and FM_{ct} that were the numerator of the numerator of the numerator.
- 3. Conversely, because the data are a snapshot of the national register as of July 2017, the variable FA_{ct} may overestimate family ties the further we move away from 2017. For example, suppose lawyers tend to join larger law firms over time. This would increase the chance of sharing a family name with a colleague and identify family ties as of July 2017 that did not exist when a lawyer entered the register.

These considerations show that our proxy for family ties may be affected by a mechanical trend that can be positive or negative depending on whether the effects in 1 and 2 prevail over the effect in 3 or not. Under some mild assumptions, FA_{ct} and FM_{ct} measure the variation in entrants with family ties at the time of the reform. In particular, assuming that the mechanical attrition is continuous at the time of the reform and conditional on a correct specification of the time trend, the break in the trend of entrants with family ties is the effect of the reform. Consequently, it is crucial to specify the time trend properly in the empirical model. In Section A we check different degrees of the time trend, allowing it to be different before and after the reform. Overall, results are significant and very similar to the main estimates in Table 2.

- N. 1322 Judicial efficiency and bank credit to firms, by Giacomo Rodano (March 2021).
- N. 1323 Unconventional monetary policies and expectations on economic variables, by Alessio Anzuini and Luca Rossi (March 2021).
- N. 1324 *Modeling and forecasting macroeconomic dowside risk*, by Davide Delle Monache, Andrea De Polis and Ivan Petrella (March 2021).
- N. 1325 Foreclosures and house prices, by Michele Loberto (March 2021).
- N. 1326 *inancial structure and bank relationships of Italian multinational firms*, by Raffaello Bronzini, Alessio D'Ignazio and Davide Revelli (March 2021).
- N.1327 Foreign investors and target firms' financial structure: cavalry or lucusts?, by Lorenzo Bencivelli and Beniamino Pisicoli (April 2021).
- N. 1328 Board composition and performance of state-owned enterprises: quasi experimental evidence, by Audinga Baltrunaite, Mario Cannella, Sauro Mocetti and Giacomo Roma (April 2021).
- N. 1329 Can internet banking affect households' participation in financial markets and financial awareness?, by Valentina Michelangeli and Eliana Viviano (April 2021).
- N. 1330 (In)Efficient separations, firing costs and temporary contracts, by Andrea Gerali, Elisa Guglielminetti and Danilo Liberati (April 2021).
- N. 1331 *The catalytic role of IMF programs*, by Claudia Maurini and Alessandro Schiavone (April 2021).
- N.1332 Dating the euro area business cycle: an evaluation, by Claudia Pacella (April 2021).
- N. 1333 Population aging, relative prices and capital flows across the globe, by Andrea Papetti (April 2021).
- N.1334 What drives investors to chase returns?, by Jonathan Huntley, Valentina Michelangeli and Felix Reichling (April 2021).
- N. 1335 Managerial talent and managerial practices: are they complements?, by Audinga Baltrunaite, Giulia Bovini and Sauro Mocetti (April 2021).
- N.1336 Volatility bursts: a discrete-time option model with multiple volatility components, by Francesca Lilla (June 2021).
- N.1337 A liquidity risk early warning indicator for Italian banks: a machine learning approach, by Maria Ludovica Drudi and Stefano Nobili (June 2021).
- N.1338 Macroprudential policy analysis via an agent based model of the real estate sector, by Gennaro Catapano, Francesco Franceschi, Michele Loberto and Valentina Michelangeli (June 2021).
- N.1339 Optimal robust monetary policy with parameters and output gap uncertainty, by Adriana Grasso and Guido Traficante (June 2021).
- N.1340 Information or persuasion in the mortgage market: the role of brand names, by Agnese Carella and Valentina Michelangeli (June 2021).
- N.1341 *The grocery trolley race in times of Covid-19: evidence from Italy*, by Emanuela Ciapanna and Gabriele Rovigatti (June 2021).
- N.1342 When the panic broke out: COVID-19 and investment funds' portfolio rebalancing around the world, by Massimiliano Affinito and Raffaele Santioni (July 2021).
- N.1343 *Firm-bank linkages and optimal policies in a lockdown*, by Anatoli Segura and Alonso Villacorta (July 2021).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., M. CIOFFI and P. TOMMASINO, *Legislators' behaviour and electoral rules: evidence from an Italian reform,* European Journal of Political Economy, v. 59, pp. 423-444, WP 1135 (September 2017).
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, v. 15, 4, pp. 55-80, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, *Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis,* Italian Economic Journal / Rivista italiana degli economisti, v. 5, 1, pp. 17-37, **WP 1086 (October 2016).**
- BASSO G., F. D'AMURI and G. PERI, *Immigrants, labor market dynamics and adjustment to shocks in the euro area*, IMF Economic Review, v. 67, 3, pp. 528-572, WP 1195 (November 2018).
- BATINI N., G. MELINA and S. VILLA, *Fiscal buffers, private debt, and recession: the good, the bad and the ugly,* Journal of Macroeconomics, v. 62, **WP 1186 (July 2018).**
- BURLON L., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of an open-ended asset purchase programme,* Journal of Policy Modeling, v. 41, 6, pp. 1144-1159, WP 1185 (July 2018).
- BUSETTI F. and M. CAIVANO, *Low frequency drivers of the real interest rate: empirical evidence for advanced economies*, International Finance, v. 22, 2, pp. 171-185, **WP 1132 (September 2017).**
- CAPPELLETTI G., G. GUAZZAROTTI and P. TOMMASINO, *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, Fiscal Studies, v. 40, 2, pp. 211-237, **WP 938 (November 2013).**
- CARDANI R., A. PACCAGNINI and S. VILLA, *Forecasting with instabilities: an application to DSGE models with financial frictions*, Journal of Macroeconomics, v. 61, WP 1234 (September 2019).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, *Fiscal consolidation by intergovernmental transfers cuts? The unpleasant effect on expenditure arrears,* Economic Modelling, v. 77, pp. 266-275, WP 1076 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, Regional Science and Urban Economics, v. 75, pp. 1-21, WP 1112 (April 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, WP 985 (November 2014).
- CIAPANNA E. and M. TABOGA, *Bayesian analysis of coefficient instability in dynamic regressions*, Econometrics, MDPI, Open Access Journal, v. 7, 3, pp.1-32, WP 836 (November 2011).
- COLETTA M., R. DE BONIS and S. PIERMATTEI, *Household debt in OECD countries: the role of supply-side and demand-side factors,* Social Indicators Research, v. 143, 3, pp. 1185–1217, WP 989 (November 2014).
- COVA P., P. PAGANO and M. PISANI, *Domestic and international effects of the Eurosystem Expanded Asset Purchase Programme,* IMF Economic Review, v. 67, 2, pp. 315-348, WP 1036 (October 2015).
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, v. 23, 8. pp. 3457-2482, **WP 1174 (April 2018).**
- FERRERO G., M. GROSS and S. NERI, *On secular stagnation and low interest rates: demography matters,* International Finance, v. 22, 3, pp. 262-278, **WP 1137 (September 2017).**
- FOA G., L. GAMBACORTA, L. GUISO and P. E. MISTRULLI, *The supply side of household finance*, Review of Financial Studies, v.32, 10, pp. 3762-3798, **WP 1044 (November 2015).**
- GERALI A. and S. NERI, *Natural rates across the Atlantic,* Journal of Macroeconomics, v. 62, article 103019, WP 1140 (September 2017).
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, v. 78, pp. 118-133, WP 1167 (March 2018).
- GOMELLINI M., D. PELLEGRINO and F. GIFFONI, *Human capital and urban growth in Italy*, 1981-2001, Review of Urban & Regional Development Studies, v. 31, 2, pp. 77-101, WP 1127 (July 2017).
- LIBERATI D. and M. LOBERTO, *Taxation and housing markets with search frictions*, Journal of Housing Economics, v. 46, article 101632, WP 1105 (March 2017).
- MAGRI S., Are lenders using risk-based pricing in the Italian consumer loan market? The effect of the 2008 crisis, Journal of Credit Risk, v. 15, 1, pp. 27-65, WP 1164 (January 2018).
- MERCATANTI A., T. MAKINEN and A. SILVESTRINI, *The role of financial factors for european corporate investment,* Journal of International Money and Finance, v. 96, pp. 246-258, WP 1148 (October 2017).
- MIGLIETTA A., C. PICILLO and M. PIETRUNTI, *The impact of margin policies on the Italian repo market*, The North American Journal of Economics and Finance, v. 50, **WP 1028 (October 2015).**

- MONTEFORTE L. and V. RAPONI, *Short-term forecasts of economic activity: are fortnightly factors useful?*, Journal of Forecasting, v. 38, 3, pp. 207-221, WP 1177 (June 2018).
- NERI S. and A. NOTARPIETRO, Collateral constraints, the zero lower bound, and the debt-deflation mechanism, Economics Letters, v. 174, pp. 144-148, WP 1040 (November 2015).
- PANCRAZI R. and M. PIETRUNTI, *Natural expectations and home equity extraction*, Journal of Housing Economics, v. 46, 4, WP 984 (November 2014).
- PEREDA FERNANDEZ S., *Teachers and cheaters. Just an anagram?*, Journal of Human Capital, v. 13, 4, pp. 635-669, WP 1047 (January 2016).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, v. 23, 2, pp. 590-624, WP 871 (July 2012).

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times*, R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, *Public debt expansions and the dynamics of the household borrowing constraint*, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities,* Journal of Business & Economic Statistics, , v. 38, 4, pp. 826-838, WP 974 (October 2014).
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, v. 37, 1, pp. 134-166, WP 1178 (June 2018)
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, **WP 1201 (November 2018).**
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy*, Journal of International Economics, v. 126, article 103328, **WP 1032** (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, *Services trade and credit frictions: evidence with matched bank firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, v. 35, 3, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome*, Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis,* Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, WP 1179 (June 2018).
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, **WP 1116 (June 2017).**
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises,* Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, **WP 1299 (October 2020).**
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, **WP 1232 (July 2019).**
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case,* Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).

- PERICOLI M., On risk factors of the stock-bond correlation, International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, WP 1059 (March 2016).
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, WP 1072 (July 2016).

2021

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019).**
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- BARONE G., F. DAVID, G. DE BLASIO and S.MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, Macroeconomic effects of non-standard monetary policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (Oct 2019).
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, WP 1187 (September 2018).
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, *Domestic and global determinants of inflation: evidence from expectile regression*, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1324 (March 2021).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit*, Journal of Regional Science, v. 61, 3, pp. 597-622, WP 1222 (June 2019).
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, *Secular stagnation, R&D, public investment and monetary policy: a global-model perspective,* Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).
- DEL PRETE S. and M. L. STEFANI, Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, WP 1014 (June 2015).
- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area*, Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds*, The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**

- LOSCHIAVO D., *Household debt and income inequality: evidence from Italian survey data*, Review of Income and Wealth. v. 67, 1, pp. 61-103, WP 1095 (January 2017).
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

FORTHCOMING

- ALBANESE G., G. DE BLASIO and A. LOCATELLI, *Does EU regional policy promote local TFP growth? Evidence* from the Italian Mezzogiorno, Papers in Regional Science, **WP 1253 (December 2019).**
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, WP 1197 (November 2018).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, WP 1295 (September 2020).
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, **WP 1133 (September 2017).**
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, WP 1269 (March 2020).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, WP 1288 (July 2020).
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, *Monetary Policy strategies in the new normal: a model*based analysis for the Euro Area, Journal of Macroeconomics, v. 70, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, WP 1224 (June 2019).
- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, **WP 1119 (June 2017).**
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, **WP 1101 (February 2017).**
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets:* evidence from the mortgage market, Journal of Financial Economics, **WP 1252 (December 2019).**
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers*, Journal of Money, Credit and Banking, **WP 1277 (June 2020).**
- LOBERTO M, *Foreclosures and house prices*, Italian Economic Journal / Rivista italiana degli economisti, WP 1325 (March 2021).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, **WP 1171 (April 2018).**
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, WP 1182 (July 2018).
- PAPETTI A., Demographics and the natural real interest rate: historical and projected paths for the Euro Area, Journal of Economic Dynamics and Control, WP 1306 (November 2020).
- PERICOLI M. and M. TABOGA, Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models, Journal of Financial Econometrics, WP 1189 (September 2018).
- PIETRUNTI M. and F. M. SIGNORETTI, Unconventional monetary policy and household debt: the role of cashflow effects, Journal of Macroeconomics, v. 64, WP 1142 (October 2017).
- SCHIVARDI F., E. SETTE and G. TABELLINI, Credit misallocation during the European financial crisis, Economic Journal, WP 1139 (September 2017).
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, WP 1243 (November 2019).